Developing decision support tools incorporating personalised predictions of likely visual benefit versus harm for cataract surgery: research programme

John M Sparrow,^{1,2*} Mariusz Grzeda,^{1,2} Andrew Frost,³ Christopher Liu,^{4,5,6} Robert L Johnston,^{7†} Peter Scanlon,⁷ Christalla Pithara,^{2,8} Daisy Elliott,² Jenny Donovan,² Natalie Joseph-Williams,⁹ Daniella Holland-Hart,⁹ Paul HJ Donachie,^{7,10} Padraig Dixon,² Rebecca Kandiyali,² Hazel Taylor,¹¹ Katie Breheny,² Jonathan Sterne,² William Hollingworth,² David Evans,¹² Fiona Fox,¹³ Sofia Theodoropoulou,¹ Rachael Hughes,² Matthew Quinn,⁵ Daniel Gray,¹⁴ Larry Benjamin,¹⁵ Abi Loose,¹ Lara Edwards,^{1,8} Pippa Craggs,¹ Frances Paget,¹ Ketan Kapoor¹ and Jason Searle¹

- ¹Ophthalmology, Bristol Eye Hospital, University Hospitals Bristol and Weston NHS Foundation Trust, Bristol, UK
- ²Population Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK
- ³Ophthalmology, Torbay Hospital, Torbay and South Devon NHS Foundation Trust, Torquay, UK
- ⁴Ophthalmology, Sussex Eye Hospital, Brighton and Sussex University Hospitals NHS Foundation Trust, Brighton, UK
- ⁵Ophthalmology, Brighton and Sussex Medical School, University of Sussex, Brighton, UK ⁶Tongdean Eye Clinic, Hove, UK
- ⁷Ophthalmology, Cheltenham General Hospital, Gloucestershire Hospitals NHS Foundation Trust, Cheltenham, UK
- ⁸Ophthalmology, National Institute for Health and Care Research Collaboration for Leadership in Applied Health Research and Care West, University Hospitals Bristol NHS Foundation Trust, Bristol, UK
- ⁹Ophthalmology, Cardiff University School of Medicine, University Hospital of Wales, Cardiff, UK
- ¹⁰National Ophthalmology Database, The Royal College of Ophthalmologists, London, UK
- ¹¹Research and Innovation, University Hospitals Bristol NHS Foundation Trust, Bristol, UK

 ¹²Health and Social Sciences, University of the West of England, Bristol, UK
¹³Translational Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK
¹⁴Ophthalmology, School of Social Science, Cardiff University, Cardiff, UK
¹⁵Ophthalmology, Stoke Mandeville Hospital, Buckinghamshire Healthcare NHS Trust, Aylesbury, UK

*Corresponding author John.Sparrow@doctors.org.uk †In memoriam

Declared competing interests of authors: Robert L Johnston (deceased) was a director of the Electronic Medical Records company from which data on cataract surgery were extracted for analyses in work package 2. Paul HJ Donachie is the statistician for the Royal College of Ophthalmologists' National Ophthalmology Database and reports grants from the Healthcare Quality Improvement Partnership outside the submitted work. Jenny Donovan declares membership of National Institute for Health and Care Research (NIHR) Rapid Trials and Health Technology Assessment (HTA) Committee Boards. Jonathan Sterne was a member of the NIHR HTA Clinical Trials and Evaluation Board from 2012 to 2015. William Hollingworth was a member of the NIHR HTA Clinical Evaluation and Trials funding committee between 2016 and 2021.

Published October 2022 DOI: 10.3310/BAGA4188

Plain English summary

Predict-CAT and Involve-CAT Programme Grants for Applied Research 2022; Vol. 10: No. 9 DOI: 10.3310/BAGA4188

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Decisions about whether to opt for cataract surgery right away or hold off are mostly based on a patient's ability to read letters on a test chart, rather than on their real-world experience of their vision. This can mean that patients are sent for surgery before they really need it or (more commonly) that they have to wait longer, that is until their vision is deemed poor enough. For the decision-making process to be more patient friendly, we developed and tested a series of decision support 'tools' that give individual patients and their doctors a better understanding of how cataracts affect everyday vision.

To enable patients to express how their vision affects them day to day, we developed a short questionnaire, Cataract Patient-Reported Outcome Measure, five items (Cat-PROM5), which has been shown to work well. Cat-PROM5 can be completed by patients before they see their eye doctor and can then be considered alongside the letter test chart results. Cat-PROM5 can also be completed after surgery, which is useful for doctors to assess how much a patient has benefited from surgery. Furthermore, our analysis allows the eye doctor to predict how good a patient's vision is likely to be after the surgery, should they decide to go ahead with it.

People considering an operation usually also want to know what the risk is of things turning out badly. We have, therefore, developed a way to predict the risk of a complication occurring during the surgery and of loss of vision following the surgery, based on things like age and eye health.

Using these predictions, we created a cataract decision aid that also includes general cataract surgery information and frequently asked questions. This was tested and most of the patients and doctors thought it was useful, although the cataract decision aid and the way it is used could still be improved.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

Programme Grants for Applied Research (PGfAR) was launched in 2013 and is indexed by Europe PMC, NCBI Bookshelf, DOAJ, Ulrichsweb[™] (ProQuest LLC, Ann Arbor, MI, USA) and Scopus[®] (Elsevier, Amsterdam, Netherlands).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar.

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health and Care Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0611-20013. The contractual start date was in December 2013. The final report began editorial review in November 2019 and was accepted for publication in October 2021. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2022 Sparrow *et al.* This work was produced by Sparrow *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk